Author's response to reviews

Title: Quantification of the energy gap in young overweight children. The PIAMA birth cohort study

Authors:

Saskia W van den Berg (saskia.van.den.berg@rivm.nl)
Jolanda MA Boer (jolanda.boer@rivm.nl)
Salome Scholtens (s.scholtens@epi.umcg.nl)
Johan C de Jongste (j.c.dejongste@erasmusmc.nl)
Bert Brunekreef (B.Brunekreef@UU.NL)
Henriette A Smit (h.a.smit@umcutrecht.nl)
Alet H Wijga (alet.wijga@rivm.nl)

Version: 3 Date: 4 May 2011

Author's response to reviews: see over
Major revisions (we require the author to make these changes)

Authors' contributions: Please indicate clearly that all authors read and approved the final manuscript. Done

References: Please make sure references are cited in numerical order. It seems that a mistake has been made when first referencing 4 and 5. (Please check your references carefully, it seems that some numbers have not been included within the square brackets). Done

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Line numbers: Please remove these. Done

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript. Done

Tables: please note that we are unable to correctly display merged cells where the merged cell crosses rows : please re-layout your table without these merged elements Done

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures I have tried to do this, but I did not succeed

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Done, I have corrected some typographical errors